Author’s response to reviews

Title: Assessment of long-term sickness absence: content and face validity of a new questionnaire based on qualitative data from nominal groups

Authors:

Kaat Goorts (kaat.goorts@kuleuven.be)
Charlotte Vanovenberghe (charlotte.vanovenberghe@kuleuven.be)
Charlotte Lambreghts (charlotte.lambreghts@kuleuven.be)
Eline Bruneel (eline.bruneel@vlaamspatientenplatform.be)
Dorina Rusu (Dorina.rusu@ulg.ac.be)
Marc Du Bois (marc.dubois@kuleuven.be)
Sofie Vandenbroeck (sofie.vandenbroeck@kuleuven.be)
Lode Godderis (lode.godderis@kuleuven.be)

Version: 1 Date: 08 Oct 2019

Author’s response to reviews:

Dear Editor,

We did not include any identifiable information in our manuscript. Therefore, I've changed the statement into "No additional informed consent was obtained from participants because no identifying information has been included in this article."